Author's response to reviews

Title: Perceived body image in men and women with type 2 diabetes mellitus: correlation of body mass index with the figure rating scale

Authors:

- Harold E Bays (hbaysmd@aol.com)
- Debbra D Bazata (debbazata@hotmail.com)
- Kathleen M Fox (kathyfox@comcast.net)
- Susan Grandy (susan.grandy@astrazeneca.com)
- James R Gavin III (jrgavin3@yahoo.com)
- SHIELD Study Group (none@yahoo.com)

Version: 7 Date: 8 December 2009

Author's response to reviews:

Minor revisions to punctuate and spelling in the text.